

GOPEN ACCESS

Citation: Siudak D (2022) The effect of selforganizing map architecture based on the value migration network centrality measures on stock return. Evidence from the US market. PLoS ONE 17(11): e0276567. https://doi.org/10.1371/journal. pone.0276567

Editor: Vygintas Gontis, Vilnius University, LITHUANIA

Received: May 30, 2022

Accepted: October 8, 2022

Published: November 1, 2022

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pone.0276567

Copyright: © 2022 Dariusz Siudak. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the paper and its <u>Supporting Information</u> files.

RESEARCH ARTICLE

The effect of self-organizing map architecture based on the value migration network centrality measures on stock return. Evidence from the US market

Dariusz Siudak *

Institute of Management, Lodz University of Technology, Lodz, Poland

* dariusz.siudak@p.lodz.pl

Abstract

Complex financial systems are the subject of current research interest. The notion of complex network is used for understanding the value migration process. Based on the stock data of 498 companies listed in the S&P500, the value migration network has been constructed using the MST-Pathfinder filtering network approach. The analysis covered 471 companies included in the largest component of VMN. Three methods: (i) complex networks; (ii) artificial neural networks and (iii) MARS regression, are developed to determine the effect of network centrality measures and rate of return on shares. A network-based data mining analysis has revealed that the topological position in the value migration network has a pronounced impact on the stock's returns.

1. Introduction

The value migration (VM) is the result of investors' seeking for effective capital allocation, increasing the stock's returns commensurate to the level of risk [1]. The knowledge of how value migrates between companies is the basis of value-based management in the considered set of enterprises. Value is a measure of the economic efficiency of a company. Since VM is the result of an evaluation of the company value carried out on the financial market, the proper measurement of value migration is an objective measure of the effectiveness of management in terms of adequate decision-making leading to increases in company value. An analysis of the VM processes enables effective value-based management and, at the same time, leads to continual reference to other companies in the considered set of enterprises. The value migration analysis conducted at various levels of aggregation allows to evaluate the effectiveness of capital allocation by investors; that is, an allocation based on expected return and estimated risk.

The VM process can be regarded as an evolving complex financial system consisting of many feedbacks. In this work, I have analyzed the VM process on the stock market through the application of complex networks. According to previous researches, complex networks are a powerful and widely used approach in econophysics field [2–10] to study the relationships among financial assets [11], where the financial system can be represented in a natural way as

Funding: The author received no specific funding for this work.

Competing interests: The authors have declared that no competing interests exist.

assets are vertices and the connections between them are links. A network-based data mining approach enables to reveal hidden information and relationships between assets that potentially affect market functioning. The network patterns embedded in the value fluctuations shed some new light on the corporate's share in the value fluctuation in the set of companies under consideration, i.e., entities listed on the stock exchange, and provide a deeper understanding of the VM process.

The main utilization of the correlation-based networks approach is to convert the multidimensional relationship matrix of the financial market into its sparse depiction. The stock correlation network is a subset of financial network that provides a deeper understanding of stock return time series [12–20], better predicts stock market behavior [21–27] and plays a significant role in portfolio optimalization [28–35], risk assessment [36–39], asset allocation [40, 41]. In other words, correlation-based networks are a useful approach in economic decision-making [42] and can be regarded as the methodological basis of portfolio theory leading to efficient risk management.

It is important to note that many studies have found that the degree distribution of the cross-correlation network follows a power-law model [26, 43–56], and displays the non-fractal property [26, 54]. Furthermore, the degree distribution in the stock network based on mutual information [57] and Engle-Granger cointegration test [58] are also fitted to a power-law model.

The value migration network (VMN) is a network built on the basis of negative correlations between returns of daily stock price and the relative changes in the firms' market capitalization. In the VMN, companies (stocks) are represented as nodes, and the value flows between stocks are modeled as edges. It has been demonstrated that in- and out-degree distribution in the VMN obeys a power-law [59]. This implies that a relatively few hub-like assets strongly influence the fluctuations of the rest stock prices and corporate values in the entire stock market. As pointed out by Siudak [59], the VMN shows a disassortative behavior. In this light, only a few highly connected assets synchronize flows of value on the stock market in two directions, the inflow and outflow of value. However, it has been revealed [40] that the topological properties of financial networks vary at diverse time scales.

Dependency structures as well as the topological position of assets in the financial filtered network influence the rate of return on shares. Some studies in this area have recently discussed which stocks are superior-central or peripheral. Previous studies provide mixed results regarding the direction of the impact of the company's centrality in the cross-correlation network on stock returns. Some studies provide evidence of a positive relationship [36, 60-64], pointing out that central stocks are preferred, while some—of a negative one [65], indicating that it is better to invest in peripheries. Further research [66], using the minimum spanning trees based on linear correlation and mutual information, pointed out differentiated results, where the dominance of the stock type-central or peripheral-in terms of the impact on stock return depends on the assumed analysis period. On the one hand, due to the observed powerlaw behavior of degree distribution in the stock correlation network, vertices with many edges to other nodes (hubs) are preferentially located in the center of the network. These highly correlated stocks can drive the movements of the peripheral stock prices and the rates of return of central stocks prevail over the peripheral stocks. On the other hand, stocks lying in the networks' peripheral area have a more diversified structure and are less susceptible to uncontrolled stock price movements when the market is volatile. This applies to assets represented as a set of stocks across the entire stock market.

Other studies have focused on the network structure and its impact on the performance of the selected portfolio, where the risk-return ratio plays a crucial role and makes the analysis more complex. It is worth pointing out that assets located in different parts of the financial

filtered networks, especially peripheries, reveal different quotation patterns on the stock market, which can be beneficial to constructing an effectively diversified portfolio. Taking the return-risk relation into consideration, it has been revealed that peripheral portfolios dominate over central portfolios [32, 41, 67, 68]. The results of peripheral assets are superior to those of central ones, because portfolios composed of peripheral stocks have a more diversified structure and achieve a lower level of risk [66, 69, 70]. In other words, a more diversified portfolio outperforms central portfolio under return-risk relation. Further study using a clustering approach has shown that portfolio optimization can be performed using either peripheral stocks or central stocks depending on the stock market condition in the selection and investment horizon [71]. It should be stressed that network stability is an important element in portfolio construction [72].

However, the aforementioned studies have only concerned the correlation-based network approach, where the cross-correlation of log-return of stock price has been investigated. In this study, I consider the value migration network that is partly based on the Pearson correlation coefficients among assets, where only negative correlation coefficients are considered (see details in Section 2). A key aim of this work is to study the relationship between the topological position in the VMN and the annualized rate of return on shares in the S&P 500. Specifically, I examine the following research question: Does the company's topological position in the VMN determine the effect on individual stock return? The search for the economic factors influencing stock returns in the highly volatile conditions of the financial market is an open and up-to-date research area.

To the best of my knowledge, no previous work has endeavored to employ three powerful data mining methods in one study: 1) complex network, 2) artificial neural networks (ANN), and 3) multivariate adaptive regression splines (MARS). This work shows that the topological position in the value migration network, measured using a self-organizing features map, plays a crucial role in determining the stock's return. Based on the empirical analysis, it has been found that the company's topological position within the VMN has an impact on the annualized rate of return on shares. Specifically, centrality measures derived from the directed network, expressing the value flow pattern, are an appropriate indicator of annualized stock return. However, the direction of the impact of individual centrality measures is not specified; instead, a self-organizing map (SOM) of value migration based on the central position of the enterprise in the VMN has been created.

The main contribution of this work is twofold. First, I have produced a network illustrating the VM process. The network approach is important due to the possibility of extensive application of complex networks in the research process. The network dimension is part of a wide range of financial market research, extending the analysis to include the element of fluctuating company value.

Based on the measures of centrality of the value migration network, the SOM was obtained using the Kohonen network. Algorithm used in the unsupervised competitive learning process allows to detect hidden patterns of the VMN structure. This goal is accomplished without specifying the output signal defining the expected network reaction. The use of artificial neural networks enables the detection of the hidden topology of the value migration network.

Secondly, I employ MARS regression models to examine the impact of a self-organizing map in terms of the centrality position in the VMN on the annualized stock's returns. This network-based data mining analysis has the benefit of extending the framework of value-based management theory. My findings shed new insights on the VM process observed on the stock market and its impact on stock return. A better understanding of the value fluctuation patterns among assets in the network approach can be applied to network-based portfolio selection and risk management. The remaining part of this paper is structured as follows. Section 2 explains the applied methodology. Section 3 briefly presents the empirical data. Section 4 includes the empirical results, and finally Section 5 provides a few conclusions.

2. Materials and methods

In this section, the research methodology is presented. The specified aim of the study has been performed employing a three-stage hybrid modeling procedure, an integration complex network and artificial neutral networks approaches with the multivariate adaptive regression splines technique. The diagram of the research process is shown in Fig 1.

First, I constructed the VMN based on i) negative Pearson correlation coefficients, used for identification connections between assets, and ii) the relative changes in market capitalization of each stock utilized to form the final direction of the relationship in the directed graph. The appropriate network construction procedure is presented in Section 2.1. The created value migration network is the basis for generating variables to search for the consequences of the network position of the enterprise and the network structure. Centrality measures (in- and out-degree; entropy; HITS centrality–see details in Section 2.5), taking into account the multifaceted position of the company in the network, were adopted as network variables.

Then I employed these centrality measures as the input variables in the input layer of SOM to detect the VMN topology based on the Kohonen network (Section 2.2). This topology was used to classify companies into disjoint groups taking into consideration specified neurons, using the entire data set (training, test, and validation set). The output layer of the Kohonen network consists of clusters with assigned companies due to the similarity according to the centrality measures. Then I sorted the groups (neurons) in a non-decreasing order according



https://doi.org/10.1371/journal.pone.0276567.g001

to the average annualized log-rate of return on shares (ARRS) and assigned ranks also in a non-decreasing manner. As a result, an independent variable (clusters of centrality) is created. Finally, MARS regression was estimated to analyze relationships between the firm's structural position in the value migration network and the rate of return on shares (see details in Section 2.3). The dependent variable and independent variables used in the MARS regression are discussed in Sections 2.4 and 2.5, respectively.

2.1. Value migration network construction

The main issue is the construction of the VMN. Mantegna [12] proposed the minimal spanning tree (MST) method, the shortest tree linking all vertices in a graph, that has been a common and functional technique to filter out the information noise from the financial network, and has been widely utilized to examine stock return time series [73–81]. The advantage of MST is the simplification of network complexity [82] and effective compression of all information and investor expectations [83] by pruning the network consisting of N(N-1)/2 links to find the N-1 most relevant edges that connect all N vertices without loops or cycles. The MST permits the tree structure, whereas other filtering methods like the Planar Maximally Filtered Graph (PMFG), asset graph, or the winner-takes-all allow to form loops. For this reason, the MST approach is the most appropriate for visualizing the VM process as a graph.

The robustness problem regarding the non-uniqueness of the MST should be emphasized when the system contains more than one MSTs [84]. The MST is unique when all link weights between each pair of nodes are dissimilar [85], which can only be fulfilled when *N* is large and $T \gg N$ [86], where *N* is the number of nodes and *T* is the length of time series. To preserve the uniqueness of the minimum spanning tree, the MST-based Pathfinder approach [87] has been applied to construct the VMN. This PFNET algorithm parametrized with ($r = \infty$ and q = N-1) directly prunes a given network as the union of all MSTs possibly existing in the original network is extracted. In other words, Pathfinder directly merges all edges with reference to the separately existing MSTs of the original network that is equivalent to the PFNET (∞ ; *N*-1). The following steps based on the procedure presented in Ref. [59] are performed to build the value migration network:

- 1. Individual time series related to the company are selected.
- 2. Based on the closing stock-price, the *N* x *N* symmetric matrix of the Pearson correlation coefficient **C** between log-returns of pairs of stocks is determined:

$$c_{ij} = \frac{\langle r_i r_j \rangle - \langle r_i \rangle \langle r_j \rangle}{\sqrt{(\langle r_i^2 \rangle - \langle r_i \rangle^2)(\langle r_j^2 \rangle - \langle r_j \rangle^2)}}$$
(1)

where

$$\langle r_i \rangle = \frac{1}{T} \sum_{t=1}^{T} r_i(t)$$
 (2)

$$r_{i} = \ln P_{i}(t) - \ln P_{i}(t-1)$$
(3)

for i, j = 1, 2, ..., N and t = 1, 2, ..., T.

3. The cross-correlation matrix \mathbf{C} is pruned to the $N \ge N$ matrix form

$$C = [c_{ij}'] \tag{4}$$

where the only negative correlation coefficients is considered as the following equation:

$$c'_{ij} = \begin{cases} c_{ij} \text{ if } c_{ij} < 0\\ 0 \text{ otherwise} \end{cases}$$
(5)

4. The yearly logarithmic return of the firm's market capitalization is calculated:

$$R_{MV_i}(t) = \ln MV_i(t) - \ln MV_i(t-1); \text{ for } i = 1, 2, \dots, N$$
(6)

where $MV_i(t)$ -market capitalization of asset *i* at time *t*; *t*-end of the year; and the return of market capitalization for entire market is computed:

$$R_{\sum MV_i(t)} = \ln \sum_{i=1}^{N} MV_i(t) - \ln \sum_{i=1}^{N} MV_i(t-1)$$
(7)

The annual window is utilized to diminish the effects of short-term fluctuations in the company value.

- 5. The specific threshold value $\theta = 0.6667$ is set.
- 6. Each stock is assigned to one of the two possible phases, (1) inflow or (2) outflow, by means of the following formulas: inflow phase:

$$\text{if } R_{MV_i}(t) \ge \theta R_{\sum MV_i(t)} \tag{8}$$

and outflow phase:

$$\text{if } R_{MV_i}(t) < \theta R_{\sum MV_i(t)} \tag{9}$$

7. Based on the reduced cross-correlation matrix **C**', the polytomous variable x_{ij} is defined as follows:

$$x_{ij} = \begin{cases} 1 \text{ if } i \in \text{Inflow phase and } j \in \text{Outflow phase} \\ -1 \text{ if } i \in \text{Outflow phase and } j \in \text{Inflow phase} \\ 0 \text{ if otherwise} \end{cases}$$
(10)

which can be formulated in an equivalent way:

$$x_{ij} = \begin{cases} 1 \text{ if } R_{MV_i}(t) \ge \theta R_{\sum MV_i(t)} \text{ and } R_{MV_j}(t) < \theta R_{\sum MV_j(t)} \\ -1 \text{ if } R_{MV_i}(t) < \theta R_{\sum MV_i(t)} \text{ and } R_{MV_j}(t) \ge \theta R_{\sum MV_j(t)} \\ 0 \text{ if otherwise} \end{cases}$$
(11)

where

$$R_{\sum MV_i(t)} = R_{\sum MV_j(t)} \tag{12}$$

8. The symmetric matrix **A** with dimensions $N \ge N$ is constructed based on the elements c_{ij} of matrix **C'** and variable x_{ij} as follows:

$$A = [a_{ij}] = |c_{ij}' \cdot x_{ij}| \text{ for } i, j = 1, 2, \dots, N,$$
(13)

where a_{ij} is in the range [0, 1], formulating an undirected graph.

9. Based on the matrix A, a simple nonlinear transformation

$$d_{ij} = \sqrt{2(1 - a_{ij})} \text{ for } i, j, = 1, 2, \dots, N$$
 (14)

within the range $[0; \sqrt{2}]$ is executed using the metric distance that satisfies the axioms of a metric distance–a) nonnegativity; b) symmetry; c) triangular inequality, forming the $N \ge N$ symmetric distance matrix $\mathbf{D} = [d_{ij}]$.

- 10. Appling the MST-Pathfinder approach [87] in conjunction with the Kruskal's algorithm [88], the matrix **D** is pruned where the sum of all edge distances (d_{ij}) in the tree is minimized. As a result, the PFNET $(\infty, N-1)$ network is extracted as the unification of all MSTs that possibly exist in the matrix **D**, denoted by $\mathbf{T} = [t_{ii}]$.
- 11. The PFNET(∞ , *N*-1) network is then transformed to a dichotomized matrix **T**', that elements t'_{ii} are defined as

$$t'_{ij} = 1 \ if \ t_{ij} > 0 \tag{15}$$

12. In the final step, the direction of value migration is specified. An adjacency matrix **A'** is obtained by giving the direction of the network whose elements are computed using the following transform rule:

$$A = [a'_{ij}] = \begin{cases} a'_{ij} = 1\\ a'_{ji} = 0\\ a'_{ij} = 0\\ a'_{ij} = 1 \end{cases} \quad if \ t'_{ij} = t'_{ji} = 1 \ and \ x_{ij} = -1\\ a'_{ij} = 0\\ a'_{ij} = 1 \end{cases} \quad if \ t'_{ij} = t'_{ji} = 1 \ and \ x_{ij} = 1$$

$$(16)$$

where $x_{ij} = -1$ means that $i \in Outflow$ phase and $j \in Inflow$ phase $x_{ij} = 1$ means that $i \in Inflow$ phase and $j \in Outflow$ phase.

The presented procedure allows us to generate the VMN that is unweighted, directed, planar, without a cycle or self-edges, simple graph. It should be emphasized that creating a connection between a pair of stocks in the VMN is possible under two conditions: i) the occurrence of a negative Pearson correlation coefficient among pair of stocks; ii) both assets have to be assigned to opposite phases of value migration. It is not possible to form a connection between two companies if both firms are in the same value migration phase or if there is a positive correlation coefficient between them.

2.2. Kohonen network

Kohonen network is a type of artificial neutral network first proposed by Kohonen [89, 90], that has the ability to create the self-organizing map according to environmental features. Kohonen network is an unsupervised learning, self-organizing, competitive network. The self-organizing map is an automatic data-analysis approach, widely used in clustering problems

and data exploration [91]. The aim of the Kohonen network is to detect similarities among data and cluster data set into different classes using a non-linear, unsupervised, and competitive learning algorithm. In other words, similar objects in the data structure are automatically clustered and nonlinear mapping of individual classes follows a learning pattern recognized by specified neurons without determining target variables. Kohonen network attempts to learn the data structure by using the features contained in the data set in a hidden way, including the input variables without network reaction as a predefined output signal. It is used to search for the division of a given set of objects into homogeneous subsets without knowing the pattern of classes. The result of the Kohonen network's learning process is a topological map that allows to determine the importance of the relevant regions of the map formed by neurons in the topological layer of the network.

The SOM does not contain a hidden layer and consists of an input layer and an output layer, the last of which is called competitive layer or topological layer. The input layer is unidimensional and contains the number of neurons corresponding to the dimensions of the input eigenvector. The competitive layer is bidimensional, and the nodes in the output layer are the equivalent of output neurons, which are distributed in a regular two-dimensional grid, curved in the *N*-dimensional input space. These two layers–input and output–are linked in two directions where all input nodes are connected to all output nodes using the weights of the synapses. These weights can be expressed as $w_{ij}(i = 1, 2, ..., n; j = 1, 2, ..., m)$, where *n* and *m* are the number of nodes in input layer and competition layers, respectively. The more similar the input objects are, the closer to each other the corresponding neurons will be located in the space of the topological layer.

The iterative Kohonen network learning algorithm can be summarized as follows [91-93]:

- Parametrization. Setting the parameters preceding the execution of the iterative procedure:

 a topological length and width of the network defining the discrete output space; ii) a learning rate parameter; iii) a neighborhood radius parameter; iv) number of epochs.
- 2. Initialization. Selection of the connecting weights from the available set of input nodes in a random manner.
- 3. Data input. The data of sample is entered into the input layer after normalization

$$\boldsymbol{x}^{k} = (\boldsymbol{x}_{1}, \boldsymbol{x}_{1}, \dots, \boldsymbol{x}_{n}) \tag{17}$$

4. Distance computation. The Euclidean distance of connection weight between input nodes and output neurons is computed:

$$s_j = \sum_{i=1}^n (x_i^k - w_{ij})^2, (j = 1, 2, \dots, m)$$
 (18)

where x_i -input value of *i*th input node.

5. Similarity matching. Finding the best-matching neuron (winning node) with the minimum Euclidean distance:

$$c = \arg\min_{j \in \{1, 2, \dots, m\}} \{s_j\}$$

$$\tag{19}$$

6. Weight update. Adjustment of the synaptic weigh vectors of all neurons applying to the adjustment formula:

$$w_{ii}(t+1) = w_{ii}(t) + h_{c,i}(t) \cdot [x_i^k(t) - w_{ii}(t)]$$
(20)

where: $h_{c,i}(t)$ -neighborhood function centered around the winning node c:

$$h_{c,j}(t) = \eta(t) \cdot \exp\left(\frac{-r_{c,j}^{2}}{2\delta(t)^{2}}\right)$$
(21)

where $r_{c,j}$ —the geometric distance between winning neuron and *j*th node; $\delta(t)$ —neighborhood radius monotonically decreasing with the learning process; $\eta(t)$ —learning rate in the range (0; 1):

$$\eta(t) = \eta_{\min} - \frac{t}{T} \cdot (\eta_{\max} - \eta_{\min})$$
(22)

where *t*-current number of iterations; *T*-number of epochs.

7. Continued training. Input of the data of the next sample and a continued stepwise recursive procedure with step 2, until all samples in the training set have been trained and all epochs have been executed.

The research sample was divided into three sets of data in a proportion of 70% for the training set, 15% for the test set, and 15% for the validation set. The training set is employed to construct the Kohonen network, the test set-to evaluate the quality of the classification as the ability to generalize, while the validation set-to verify the applicability of the created SOM model in the network training process. The SOM quality evaluation is performed using the quantization error, which measures the mean distance between each data case and its corresponding winning neuron

$$Q_{error} = \frac{1}{N} \sum_{i=1}^{n} (x_i^k - w_{c,i})^2$$
(23)

where: $w_{c,i}$ -weight of the winning neuron corresponding to the input value of the *i*th input node.

The quantization error assesses the fit of the topology map to the data. The best result of the Kohonen network algorithm is yielded for the minimum quantization error value.

In order to determine the SOM, the following parameters were adopted: i) learning rate: Start = 0,1, End = 0,02; ii) neighborhood radius: Start = 3; End = 0; iii) number of Epochs: 200; iv) network initialization: random, Gaussian; v) topological network length and width: a combination of length/width from 3 to 6.

As a result, 10 networks were created to test the dimensions of the topological map from 3×3 to 6×6 . The projection of the self-organizing feature map is the basis for the further analysis process using the MARS regression approach.

2.3. Multivariate adaptive regression splines method

Multivariate adaptive regression splines is a non-parametric and nonlinear regression method, first developed by Friedman [94]. It is an adaptive procedure for effective modeling of complex relationships embedded in multivariate datasets. Furthermore, MARS has the ability to reliably track complex data structures across all degrees of interactions. A regression model is automatically created by considering the nonlinear interaction between variables by fitting the data into a series of spline functions. MARS regression has several advantages with respect to the ability to model complex relationships among variables, namely: (1) no prior assumptions are required for the underlying relationships between a dependent variable and a set of independent variables, as well as the distribution of these variables; (2) quantitative and qualitative explanatory variables can be included in the model; (3) the ability to flexibly model the

nonlinearity effects and predictors interactions; (4) any degree of interactions between explanatory variables is allowed; (5) unlike other regression techniques, there is no dimensionality problem.

MARS model is a nonlinear relationship between the dependent variable and the explanatory variables by means of applying a set of piece-wise regression spline functions implemented by a sequence of linear polynomial basis functions in distinct intervals of the independent variable space. The spline basis functions use two-sided truncated reflected pairs [95]:

$$(x-t)_{+} = \begin{cases} x-t, & \text{if } x > t \\ 0, & \text{if } x \le t \end{cases}$$
(24)

$$(t-x)_{+} = \begin{cases} t-x, & \text{if } x < t \\ 0, & \text{if } x \ge t \end{cases}$$
(25)

where: *t* is the knot point of basis function, index "+" means positive part of the basis function.

The slope of the regression line is expressed over an interval and changes when the two knot points is crossed. This means that the explanatory variable may enter the model with a different sign and a different value of the coefficient depending on which side of the knot point is its value.

The MARS model for dependent variable *y* with *M* spline functions can be described as [95]:

$$y = f(x) = \beta_0 + \sum_{m=1}^{M} \beta_m h_m(x)$$
(26)

where: β_0 -constant; β_m -estimated coefficients of basis functions; *M*-number of spline basis functions; $h_m(x)$ -single spline function or a product of two or more basis functions for distinct predictors (*x*) [94]:

$$h_m(x) = \prod_{k=1}^{k_m} [s_{km}(x_{\nu(k,m)} - t_{k,m})]_+$$
(27)

 k_m -number of knot points; s_{km} -+1 (-1) values that indicate right (left) sense of the associated step function; v(k,m)-label of the predictors (x); $t_{k,m}$ -knot location.

The fitting of the coefficients of spline basis functions to the data is performed using the standard OLS procedure [94]. The maximum possible number of all spline basis functions is 2Np, where *N* is the number of observations corresponding to the number of vertices in the VMN network, and *p* is the number of explanatory variables.

Then, an intensive search procedure is carried out for selecting explanatory variables and estimating the knot points for each variable. The two-stage process is performed to optimize the final form of MARS model. In the first stage, forward selection is carried out by successively adding reflected pairs of basis functions to the model. As a result, a large number of basis functions is included into the model, leading to overfitting the data. Then, the backward deletion phase procedure is initiated. The over-fit model is pruned by removing the spline functions in order of least contribution to the accuracy of the model fit based on the generalized cross-validation (GCV) criterion [96] defined as a quotient of the mean squared residual error and a penalty of the model complexity [97]:

$$GCV(M) = \frac{\frac{1}{N} \sum_{i=1}^{N} (y_i - \hat{y})^2}{\left[1 - \frac{C(\tilde{M})}{N}\right]^2}$$
(28)

where: $C(\tilde{M}) = C(M) + d \cdot M$; *d*-penalty factor; C(M)-number of parameters being fit; y_i -observed dependent variable; \hat{y} -predicted target variable.

During the backward procedure, the least significant basis functions are removed from the MARS model in terms of the measure of fit. The model reduction in the backward sequence is stopped when the minimum of the GCV(M) is attained.

Parameters for the MARS regression analysis were individually set depending on a specific model. These parameters are presented in the Section 4, where the results are presented. The variables were normalized prior to the execution of the MARS models using the following transformation function:

$$x_{ij}^{N} = \frac{x_{ij}}{\max_{i}\{x_{ij}\} - \min_{i}\{x_{ij}\}} (\max_{i}\{x_{ij}\} - \min_{i}\{x_{ij}\} \neq 0)$$
(29)

The normalized variable is in the range

$$x_{ij}^{N} = \left\langle \frac{\min_{i}\{x_{ij}\}}{\max_{i}\{x_{ij}\} - \min_{i}\{x_{ij}\}}, \frac{\max_{i}\{x_{ij}\}}{\max_{i}\{x_{ij}\} - \min_{i}\{x_{ij}\}} \right\rangle$$
(30)

and this range is constant, equal to one.

2.4. Dependent variable

The dependent variable is the annualized log-rate of return on shares

$$ARRS_{i} = \ln P_{i}(t) - \ln P_{i}(t - \Delta t)$$
(31)

for all i = 1, 2, ..., N, where *ARRS_i*-annualized rate of return on shares i; Δt -one year.

2.5. Independent variables

Three measures were used as dependent variables in the MARS regression analysis. Two of them relate to measures of VM process, and one variable is aggregated based on the centrality measures of VMN.

2.5.1. Value migration measures.

(i) Share in value migration balance (*SVMB_i*):

$$SVMB_{i} = \frac{\Delta MV_{i}}{\left|\sum_{i=1}^{N} \Delta MV_{i}\right|} \left(\sum_{i=1}^{N} \Delta MV_{i} \neq 0\right)$$
(32)

where $\sum_{i=1}^{N} \Delta M V_i$ is the balance of VM defined as the sum of the inflows and the outflows of the company's market capitalization; MV_i -market capitalization of the companies i = 1, 2, ..., N.

The flows in a company's value may compensate for one another, which in result leads to a zero balance of value migration of the considered set of corporates within one period of analysis. Hence, one needs to introduce a limiting condition:

$$\sum_{i=1}^{N} \Delta M V_i \neq 0 \tag{33}$$

If $\sum_{i=1}^{N} \Delta M V_i = 0$, then the share in migration balance, calculated separately for each company, will be undetermined. The variable share in the value migration balance (*SIVMB_i*) was adapted and reformulated from the work [98].

(ii) $MV_{i,(t-1)}$ -the market capitalization of the company *i* at the beginning of the measurement period of the VM processes.

2.5.2. Network measure.

(iii) Cluster of centrality (*CC_i*)–is an independent variable identifying the group to which the company has been qualified by the Kohonen neutral network based on centrality measures of the value migration network. This variable is the result of applying two research methods: complex network and Kohonen neutral network.

2.5.3. Centrality measures. Because network centrality has a significant role in determining the internal structure of a financial network [99], in order to assign a company to the appropriate neuron (variable: cluster of centrality), I used four centrality measures dedicated to the directed network. The applied centrality measures reflect dissimilar premises and take into account a different type of central position of enterprise in the VMN. Pozzi, Di Matteo and Aste [32] pointed out that the different sensitivity of individual centrality measures to outliers and noise justifies the combined use of these centrality indices instead of utilizing each of them in isolation. The following four centrality measures are the input variables in the input layer of the Kohonen network:

i. In-degree centrality-the portion of vertices adjacent to the node *i*

$$DC_i^{in} = \frac{k_i^{in}}{N-1} \tag{34}$$

ii. Out-degree centrality-the portion of vertices adjacent from the vertex *i*

$$DC_i^{out} = \frac{k_i^{out}}{N-1} \tag{35}$$

iii. Entropy centrality–which is related to the distribution of the probabilities that the flow stops at each of the vertices in the graph [100], defined as

$$C_{i}^{H} = \frac{-\sum_{j \in V} p_{ij} \log p_{ij}}{\log N}$$
(36)

where

$$p_{ij} = -\sum_{k=1}^{K(ij)} \sigma_k(j) \prod_{t=0}^{n(k)-1} \tau_k(v_t); \ \sigma_k(v_t)$$
(37)

is stopping probability and $\tau_k(v_t)$ denotes transfer probability.

iv. HITS centrality. The HITS approach–hyperlink-induced topic search algorithm developed by Kleinberg [101], where each vertex in a network has two types of centrality: an authority centrality and a hub centrality. HITS centrality expresses the difference between an authority centrality and a hub centrality

$$HITS_i = ac_i - h_i \tag{38}$$

where:

$$ac_i = \alpha \sum_i a'_{ij} \cdot h_j \tag{39}$$

$$h_i = \beta \sum_j a'_{ij} \cdot ac_j \tag{40}$$

 a'_{ij} -elements of the adjacency matrix \mathbf{A}' ; $ac_i(ac_j)$ -authority centrality of vertex i(j); $h_i(h_j)$ -hub centrality of vertex i(j); α, β -constant.

Authority centrality and hub centrality are in the interval $\langle 0; 1 \rangle$, which means that the HITS centrality is in the range $\langle -1; 1 \rangle$. In the VMN value can flow in or out of any company. If the value flows into the enterprise, then vertex has only in-neighbor, non-zero authority score, hub centrality equals zero, and HITS centrality has a positive value of (between 0 and 1). If the value flows out of the company, then vertex has only out-neighbor, non-zero hub centrality score, authority centrality is zero, and HITS centrality has a negative value of (between -1 and 0).

3. Data set

I used a dataset consisting of daily closure prices of 498 stocks that were continuously listed in the S&P 500 index between the end of 2018 to the end of 2019 (253 daily closing prices in each time-series). The data were obtained from Yahoo Finance [102]. In order to construct the value migration network, data on the market capitalization of companies at the end of 2018 and 2019 were used. This means that in the design of the VMN, the length of the time-window of one year was used to explore the fluctuating market value of the company. The data were retrieved from Refinite database [103]. The data used in further analyses was included in the supplementary material associated with this article (S1 Dataset).

The value migration network is shown in Fig 2. The VMN consists of 498 vertices and is a directed and unweighted network. The number of connected nodes within the largest component is 471 and the number of edges is N-1 = 470, which is characteristic of MST-based networks. The remaining 27 companies are isolated nodes, which means that their source (sink) of the inflow (outflow) of value is placed outside the analyzed set of companies. The analysis covered 471 companies included in the largest component of VMN.

All network analyses were carried out in NetMiner software [105] while Kohonen network and MARS regression were performed in Statistica software [106].

Several previous studies based on data from the US stock market have shown that one company–General Electric (GE)–is the hub vertex and occupies the central position in the correlation-based MST [12, 15, 66, 74, 75, 107, 108] and PMFG [107]. The above results have been confirmed in another study [109], in which two MST-based networks were constructed for the time series of stock returns in 1998–1999 for the 1000 and 100 largest US companies, respectively.

In this paper, the American Water Works Company (AWK) is the center of the value migration network, which has the second largest number of sources of value inflow, after Newmont Corporation (NEM); k_i^{in} is 33 and 36, respectively. General Electric is not found to be the hub, although it is in the inflow phase; $k_i^{in} = 2$. It should be noted that there are significant differences in the construction of the VMN and the cross-correlation network.

4. Results and discussion

This section reports the impact of centrality measures of the VMN on the annualized stock return. After determining the four centrality measures for the value migration network, a neutral Kohonen network was applied. The dimension of the SOM from 3 x 3 to 6 x 6 was tested and the most appropriate one was determined based on the minimum error level criterion for



Fig 2. Value migration network. Green nodes denote companies in the inflow stage; red vertices indicate firms in the outflow stage; black nodes are isolated (visualization algorithm: Fruchterman & Reingold [104]; performed in NetMiner [105]).

https://doi.org/10.1371/journal.pone.0276567.g002

the test and validation sets. The error level for the training set is less important because ANN tends to over-fit the model to the data (network overfitting) in the network learning process, hence the mean error may be close to zero. In the network assessment, the most important information is the value of the validation error obtained from the validation set. An additional test of the network model is performed using a test set to ensure that the achieved error levels for the training and validation sets are correct regardless of the mechanism of the network training process.

Of the possible 10 networks presented in Table 1, the lowest validation error level is achieved for the 3 x 4 network dimension, and the value of test error is close to the value of error obtained from the training and validation set. Because of this, as well as a lack of over-fitting or the inability to generalize, SOM with a topological layer size of 3 x 4 is used for further analysis.

The numbers of firms classified into disjoint clusters in the self-organizing map for the separate data sets are presented in Table 2 and Fig 3A. Fig 3B shows the distance map between neurons.

For the combined training, test and validation sets, a total of 8 neurons in the topological layer were obtained: (1, 1); (2,1); (3,1); (3,2); (1,3); (1,4); (2,4),(3,4). This represents the topological map of the centrality level for the entire dataset based on the value migration network. Table 3 presents the number of firms included in separate clusters, as well as the proportion and mean annualized log-rate of return on shares (mean *ARRS_i*). The clusters were ordered in

Network Id	Network*	Topological layer size	Quantization error		
			training	test	validation
1	SOM 4-9	3x3	0.00444	0.00318	0.00301
2	SOM 4-12	3x4	0.00308	0.00319	0.00225
3	SOM 4-16	4x4	0.00173	0.00347	0.00311
4	SOM 4-20	4x5	0.00132	0.00377	0.00359
5	SOM 4-25	5x5	0.00078	0.00491	0.00385
6	SOM 4-15	3x5	0.00217	0.00324	0.00297
7	SOM 4-18	3x6	0.00141	0.00367	0.00335
8	SOM 4-24	4x6	0.00113	0.00405	0.00384
9	SOM 4-30	5x6	0.00044	0.00439	0.00309
10	SOM 4-36	6x6	0.00060	0.00442	0.00333

Table 1. Errors for subsequent networks.

*—the first number indicates the number of variables, the second number is the size of the output The network selected for further analysis is highlighted in bold

https://doi.org/10.1371/journal.pone.0276567.t001

a non-descending order with respect to mean $ARRS_i$ and ranks were assigned–see the last column in Table 3. The higher the mean $ARRS_i$ in the group, the higher the rank.

Then, two regression models, which took the network structure into consideration as an independent variable, were constructed for which the *cluster of centrality* (CC_i) was assumed. The dependent variable is the annualized rate of return on shares *i* ($ARRS_i$). The explanatory variables are cluster of centrality (CC_i), share in value migration balance ($SVMB_i$), and the market capitalization of company *i* at the beginning of the measurement period of value migration processes ($MV_{(t-I)}$).

The first model was constructed on the basis of three explanatory variables. The value of the maximum number of spline basis functions is restricted to 14, the penalty factor equals 2, and no interactions between the variables were allowed. Table 4 reports the parameters and the results obtained for model 1.

MARS model contains 7 basis functions that are a single spline specified by only one predictor. The value of the overall Generalized Cross Validation (GCV) statistic is 0.0068, and the adjusted coefficient of determination (*adjusted* R^2) is 0.5856. The value of GCV criterion close to zero indicates the reliability of the model. The explanation of the variance exceeds 58.5%. It can be seen from Table 4 that all regression coefficients of the contained spline basis functions are statistically significant at a significance level of 1% or less, which implies that all variables play crucial roles in determining the annualized stock's return.

In Table 4, the regression coefficients (β) are given for each basis function. The first two basis functions BF1 and BF2 relate to the non-linear effect of the cluster of centrality in the VMN. The value of the knot point of BF1 and BF2 is 5 (the original value of 0.714 normalized knot). This means that the critical threshold of the group to which the company has been

Sample:	Training Test		Validation			Training, Test, Validation										
Cluster position	1	2	3	4	1	2	3	4	1	2	3	4	1	2	3	4
1	2	0	94	16	0	0	15	4	0	0	20	5	2	0	129	25
2	39	0	0	14	14	0	0	1	8	0	0	3	61	0	0	18
3	6	159	0	1	0	35	0	1	3	30	0	1	9	224	0	3

Table 2. Representation of companies in SOM 3 x 4 for specific datasets.

https://doi.org/10.1371/journal.pone.0276567.t002



https://doi.org/10.1371/journal.pone.0276567.g003

qualified by the Kohonen neutral network based on centrality measures of the value migration network occurs at number 5. The effect of the cluster of VMN can be diverse: (1) if the company is classified to group number 6, 7, or 8, then BF1 indicates a positive effect on the stock's return; (2) if the company is classified to group number from 1 to 4, then BF2 indicates a negative effect on the stock's return. For companies included in group 5, centrality in the VNM network does not affect the generated rate of return on shares. It turns out that it is a neuron (1, 1) containing only 2 enterprises (see Table 3).

However, taking into account three elements: (i) the form of spline basis functions–BF1: (x-t); BF2: (t-x)–(ii) the sign (+x or -x) standing next to the dependent variable and (iii) positive regression coefficient for BF1 or negative for BF2, the direct direction of changes in CC_i and $ARRS_i$ is the same, as can be summarized using appropriate notation

for BF1:

for
$$CC_i > 5$$
 and $\nearrow CC_i \longrightarrow \nearrow BF1 \longrightarrow \nearrow ARRS_i$

for
$$CC_i > 5$$
 and $\searrow CC_i \longrightarrow \bigcirc BF1 \longrightarrow \bigcirc ARRS$

for BF2:

for
$$CC_i < 5$$
 and $\nearrow CC_i \longrightarrow BF1 \longrightarrow \nearrow ARRS_i$
for $CC_i < 5$ and $\searrow CC_i \longrightarrow \nearrow BF1 \longrightarrow \searrow ARRS_i$

where \nearrow and \searrow denote the increase and decrease of the variable, respectively.

TAING J. AVCLASS VALUE OF AREA THE DRIVENUAL DEDUCTION AND THE PRODUCTING.	Table 3.	Average va	alue of ARRS	for individual	neurons and	their ordering.
--	----------	------------	--------------	----------------	-------------	-----------------

Neuron	n	Proportion	Cumulative Proportion	Mean ARRS _i	Cluster of centrality (<i>CC_i</i>)
(2,4)	18	0.038	0.038	-0.0342	1
(1,4)	25	0.053	0.091	-0.0055	2
(3,4)	3	0.006	0.097	0.0525	3
(1,3)	129	0.274	0.371	0.0572	4
(1,1)	2	0.004	0.375	0.2645	5
(3,2)	224	0.476	0.851	0.3149	6
(3,1)	9	0.019	0.87	0.3153	7
(2,1)	61	0.13	1	0.3772	8

https://doi.org/10.1371/journal.pone.0276567.t003

MODEL 1								
	Parameters	Value	Value					
Dependent varial	ble	ARRS _i						
Independent vari	ables	CC_{i} ; $SVMB_{i}$; $MV_{(t-1)}$						
Maximum numb	er of basis functions	14						
Order of interact	ion	1						
Penalty		2						
Threshold		0.0005						
Removal of irrele	want basis functions	Yes						
Number of obser	vations	471	471					
Basis function		β	t statistics	p				
	Constant	0.135 (0.011)	12.145***	0.0000				
BF1	$\max(0, CC_i - 0.714)$	0.238 (0.034)	6.927***	0.0000				
BF2	$\max(0, 0.714 - CC_i)$	-0.255 (0.032)	-7.881***	0.0000				
BF3	$max(0, SVMB_i - 0.014)$	2.747 (0.727)	3.777***	0.0002				
BF4	$\max(0, 0.014 - SVMB_i)$	-4.618 (0.492)	-9.392***	0.0000				
BF5	$\max(0, MV_{(t-1)} - 0.049)$	-0.209 (0.060)	-3.461***	0.0006				
BF6	$\max(0, 0.049 - MV_{(t-1)})$	1.228 (0.306)	4.018***	0.0001				
BF7	$max(0, SVMB_i - 0.035)$	-2.392 (0.739)	-3.237**	0.0013				
Results		Value	Value					
GCV		0.0068	0.0068					
R ²		0.5927						
Adjusted R ²		0.5856	0.5856					

Table 4. Parameters and results for model 1.

standard error in parentheses

***, ** and * denote significant levels at 0.1%, 1%, and 5%, respectively

https://doi.org/10.1371/journal.pone.0276567.t004

This means that regardless of the initial value of CC_i , companies from the cluster with a higher index (CC_i) have a higher rate of return on shares. In other words, if CC_i is greater than 5, then the greater the value of CC_i , the greater the $ARRS_i$, and vice versa. If CC_i is less than 5, the greater the CC_i value (up to the limit of $CC_i = 4$), the smaller the decrease in $ARRS_i$, and if the smaller the CC_i value, the greater the decrease in $ARRS_i$.

The basis functions BF3 and BF4 refer to the share in the value migration balance, where the critical threshold is 0.0014 (the original value of normalized knot point is 0.0141). If the share in the VM balance is greater than 0.14%, then the impact on stock return is positive and vice versa. On the other hand, BF7 indicates that if the value of $SVMB_i$ is greater than 0.347% (the original value of 0.0348 normalized knot point), then the share in the VM balance has a negative effect on the annualized stock return. Comparing the similar forms of the spline basis functions BF3 and BF7 (*x*-*t*), it can be seen that they have opposite signs for the regression coefficient–positive and negative, respectively. From the theoretical point of view, a larger share in the VM balance should contribute to achieving a higher rate of return on shares. Hence, for a spline of the form (*x*-*t*) and for the $SVMB_i$ variable, only a positive coefficient of the basis functions is justified. However, taking into account the simultaneous impact of both basis functions on the dependent variable, it should be emphasized that the BF3 function has a smaller knot point value than BF7 (0.14% vs 0.347%) as well as a greater influence as measured by the absolute value of the regression coefficient, $|\beta_{BF3}| = 2.747 > |\beta_{BF7}| = 2.392$. This means that, considering the non-linear complexity of the MARS regression model, the final effect of

the *SVMB_i* variable on the dependent variable is positive, especially when the value of the share in the VM balance is within the range (0.14% \div 0.347).

The interpretation of the pair of basis functions BF3 and BF4 is the same as for the pair of basis functions BF1 and BF2.

The last pair of piece-wise regression spline functions is BF5 and BF6, which reveal the non-linear effect of the company's market capitalization delayed by one year $(MV_{(t-1)})$ on stock's return. A negative regression coefficient for the BF5 function indicates that if the market value at the beginning of the VM measurement process is greater than 3.684E+10 USD (the original value of normalized knot point is 0.04948) then the annualized rate of return on shares is lower, while the $MV_{(t-1)}$ is greater. On the other hand, if the $MV_{(t-1)}$ is below 3.684E +10 USD, then the rate of stock return increases while the $MV_{(t-1)}$ decreases, as implied by the positive correlation coefficient of BF6 spline function and the negative sign for $MV_{(t-1)}$ in the equitation. Construction of basis functions BF5 and BF6 and the values of their coefficients indicate the inverse relationships between the $MV_{(t-1)}$ and stock's returns, which can be summarized as follows:

for BF5:

$$\begin{aligned} &\textit{for } MV_{(t-1)} > 3.684E + 10 \textit{ and } \nearrow MV_{(t-1)} \longrightarrow \nearrow BF5 \longrightarrow \ \ ARRS_i \\ &\textit{for } MV_{(t-1)} > 3.684E + 10 \textit{ and } \searrow MV_{(t-1)} \longrightarrow \ \ BF1 \longrightarrow \nearrow ARRS_i \end{aligned}$$

for BF6:

$$\begin{aligned} & \textit{for } MV_{(t-1)} < 3.684E + 10 \textit{ and } \nearrow MV_{(t-1)} \longrightarrow \ BF5 \longrightarrow \ ARRS_i \\ & \textit{for } MV_{(t-1)} < 3.684E + 10 \textit{ and } \searrow MV_{(t-1)} \longrightarrow \ \BF1 \longrightarrow \ \ARRS_i \end{aligned}$$

where \nearrow and \searrow denote the increase and decrease of the variable, respectively. The final regression equation of model 1 is as follows:

$$\begin{split} ARRS_{i} &= 0.135 + 0.238 \cdot \max(0, \ CC_{i} - 0.714) - 0.225 \cdot \max(0, \ 0.714 - CC_{i}) + 2.747 \cdot \max(0, \ SVMB_{i} - 0.014) - 4.618 \cdot \max(0, \ 0.014 - SVMB_{i}) - 0.209 \cdot \max(0, \ MV_{(t-1)} - 0.049) \ (41) \\ &+ 1.228 \cdot \max(0, \ 0.049 - MV_{(t-1)}) - 2.392 \cdot \max(0, \ SVMB_{i} - 0.035) \end{split}$$

The results obtained from model 1 imply that the company's place on the self-organizing feature map in terms of its central position in the VMN and the share in VM balance have a positive effect on the dependent variable, and the market capitalization of company delayed by one year has a negative impact. The obtained results are presented in Fig 4.

The second model contains the same explanatory variables as model 1. The maximum number of basis functions and the penalty factor are also at the same level. Unlike model 1, a second-order interactions between predictors are allowed. <u>Table 5</u> displays the parameters and MARS regression results for the second model.

From Table 5, one can find that, firstly, the overall GCV is lower compared to model 1 and equals 0.0048, and secondly, that model 2 explains the variability of the dependent variable to a greater extent (adjusted R^2 is 0.7147). The lower value of GCV and the explanation of over 71% of the variance of the annualized stock's return lead to the conclusion that model 2 is more robust compared to model 1.

The second model consists of 6 basis functions with a single spline and 4 basis functions with second-order interactions. Model 2 indicates that all independent variables play an important role in determining the annualized stock's return as all regression coefficients of



Fig 4. Graphical illustration of MARS model 1. (a) ARRS values observed versus predicted; (b) first-order term of the predictor variables CC_i and $SVMB_i$; (c) first-order term of the predictor variables CC_i and $MV_{(i-1)}$; (d) first-order term of the predictor variables $MV_{(i-1)}$; and $SVMB_i$ (color in print).

https://doi.org/10.1371/journal.pone.0276567.g004

basis functions are statistically significant with the significance level set at 5%. The results are comparable to model 1.

The first four spline basis functions (BF1, BF2, BF3, BF4) have the same nonlinear form along with the values of the knot points as their counterparts in model 1. The only differences are the values of the regression coefficients of these functions, although the signs of the coefficients are preserved. This means that the independent variables–cluster of centrality and the share in VM balance–and the dependent variable move into the same direction, which confirms the conclusions from the analysis of model 1. It should be noted that the increase in the rate of return on shares applies to companies classified to clusters 6, 7, or 8, and their share in the value migration balance is greater than 0.14%.

The last pair of piece-wise spline functions in model 2 is BF9 and BF10, related with the variable $MV_{(t-1)}$. The construction of these functions is equivalent to the forms corresponding to the pair of basis functions in model 1 –BF5 and BF6, respectively. The function of the form (*x*-*t*) is associated with a negative regression coefficient and (*t*-*x*) relates to a positive regression coefficient. The difference concerns the values of these coefficients as well as the value of the knot point. The threshold is 1.948E+11 USD (the original value of BF9 and BF10 is 0.2617) and is much higher than *t* = 3.685E+10 USD for model 1. The negative relationship between $MV_{(t-1)}$ and stock's return applies to companies whose market capitalization at the beginning of the MV process measurement is greater than 1.948E+11 USD. Companies with relatively high market values find it more difficult to generate abnormal returns on stocks. In this context, the higher knot point values of the basis functions BF9 and BF10 in model 2, compared to their model 1 counterparts, are more suitable, increasing the reliability of model 2.

Considering the basis functions with second-order interactions (BF5÷BF8), three types of interactions were found: (i) interaction between $MV_{(t-1)}$ and CC_i (BF5); (ii) interactions

Table 5. Parameters and results for model 2.

MODEL 2							
	Parameters	Value	Value				
Dependent variabl	e	ARRS _i					
Independent varia	bles	CC _i ; SVMB _i ; MV _(t-1)					
Maximum number	r of basis functions	14					
Order of interactio	n	2					
Penalty		2					
Threshold		0.0005					
Removal of irreleva	ant basis functions	Yes					
Number of observa	ations	471					
Basis function		β	t statistics	p			
	Constant	0.063 (0.018)	3.393***	0.0008			
BF1	$\max(0, CC_i - 0.714)$	0.226 (0.063)	3.595***	0.0004			
BF2	$\max(0, 0.714 - CC_i)$	-0.164 (0.028)	-5.753***	0.0000			
BF3	$max(0, SVMB_i - 0.014)$	1.011 (0.235)	4.304***	0.0000			
BF4	$\max(0, 0.014 - SVMB_i)$	-11.496 (0.679)	-16.931***	0.0000			
BF5	$\max(0, 0.072 - MV_{(t-1)}) \cdot \max(0, CC_i - 0.714)$	13.322 (1.342)	9.925***	0.0000			
BF6	$\max(0, SVMB_i - 0.064) \cdot \max(0, CC_i - 0.714)$	-1.365 (0.531)	-2.569*	0.0105			
BF7	$\max(0, 0.064 - SVMB_i) \cdot \max(0, CC_i - 0.714)$	-12.777 (1.783)	-7.165***	0.0000			
BF8	$\max(0, 0.014 - SVMB_i) \cdot \max(0, MV_{(t-1)} - 0.013)$	45.168 (3.898)	11.588***	0.0000			
BF9	$\max(0, MV_{(t-1)} - 0.262)$	-0.225 (0.075)	-2.988**	0.0030			
BF10	$\max(0, 0.262 - MV_{(t-1)})$	0.654 (0.083)	7.844***	0.0000			
Results		Value	Value				
GCV		0.0048	0.0048				
R^2		0.7214	0.7214				
Adjusted <i>R</i> ²		0.7147	0.7147				

standard error in parentheses

***, ** and * denote significant levels at 0.1%, 1%, and 5%, respectively

https://doi.org/10.1371/journal.pone.0276567.t005

between $SVMB_i$ and CC_i (BF6, BF7); (iii) interaction between $SVMB_i$ and $MV_{(t-1)}$ (BF8). Interpretation of the results based on basis functions regarding second- or higher-order interactions is quite complex. However, the inclusion of the spline basis functions BF5÷BF8 in model 2 improves the explanation of the variance of the dependent variable.

The overall regression equation for model 2 is as follows:

$$\begin{split} & ARRS_i = 0.063 + 0.226 \cdot \max(0, \ CC_i - 0.714) - 0.164 \cdot \max(0, \ 0.714 - CC_i) + 1.011 \cdot \\ & \max(0, \ SVMB_i - 0.014) - 11.496 \cdot \max(0, \ 0.014 - SVMB_i) + 13.332 \cdot \max(0, \ 0.072 - \\ & MV_{(t-1)}) \cdot \max(0, \ CC_i - 0.714) - 1.365 \cdot \max(0, \ SVMB_i - 0.064) \cdot \max(0, \ CC_i - 0.714) - \\ & 12.777 \cdot \max(0, \ 0.064 - SVMB_i) \cdot \max(0, \ CC_i - 0.714) + 45.168 \cdot \max(0, \ 0.014 - SVMB_i) \, (42) \\ & \cdot \max(0, \ MV_{(t-1)} - 0.013) - 0.225 \cdot \max(0, \ MV_{(t-1)} - 0.262) + 0.654 \cdot \max(0, \ 0.262 - \\ & MV_{(t-1)}) \end{split}$$

Graphical presentation of MARS model 2 is shown in Fig 5.

Table 6 presents the analysis of the importance of variables for models 1 and 2.

In the first model, the most important variable for the prediction of the annualized rate of return on shares is the cluster of centrality (CC_i) . This implies that the place of the enterprise





https://doi.org/10.1371/journal.pone.0276567.g005

on the self-organizing feature map of the central position in the VMN is the most dominant factor. Excluding the CC_i variable from model 1 causes the R^2 statistic to drop from 0.5927 to 0.4154, and the GCV to increase from 0.0068 to 0.0096. When the second-order interaction terms are considered in model 2, the most important variable for the prediction of the output variable is the share in value migration balance ($SVMB_i$). The importance of the CC_i variable ranks third in the hierarchical order. Ultimately, the relative importance of CC_i is over 53% of

Model 1							
Variable	Basis functions	GCV	R^2	Relative importance			
CCi	BF1, BF2	0.0096	0.4154	100.00%			
SVMB _i	BF3, BF4, BF7	0.0085	0.4685	62.80%			
MV _(t-1)	BF5, BF6	0.0071	0.5612	10.65%			
Model 2							
Variable	Basis functions	GCV	R^2	Relative importance			
CC_i	BF1, BF2, BF5, BF6, BF7	0.0067	0.5946	53.92%			
SVMB _i	BF3, BF4, BF6, BF7, BF8	0.0085	0.4685	100.00%			
$MV_{(t-1)}$	BF5, BF8, BF9, BF10	0.0071	0.5651	64.99%			

Table 6. Evaluation of the importance of the variables.

https://doi.org/10.1371/journal.pone.0276567.t006

Variable	Model 3	Model 4
Constant	-0.092 *** (-0.092)	-0.102 *** (0.015)
CC _i	0.315 *** (0.017)	0.314 *** (0.019)
SVMB _i	0.769 *** (0.124)	3.319 *** (0.629)
	-0.309 *** (0.057)	-0.054 (0.177)
$MV_{(t-1)}^*CC_i$		-0.403 (0.215)
$\overline{SVMB_i^*CC_i}$		-1.648 * (0.751)
SVMB _i *MV _(t-1)		-0.643 (0.331)
Number of observations (<i>n</i>)	471	471
F-test	161.778 ***	99.326 ***
$\overline{R^2}$	0.5096	0.5622
Adjusted <i>R</i> ²	0.5065	0.5566
Mean VIF	2.158	48.647
Max. VIF	2.760	127.798

Table 7. Linear regression analysis results.

standard error in parentheses

***, ** and * denote significant levels at 0.1%, 1%, and 5%, respectively

https://doi.org/10.1371/journal.pone.0276567.t007

the most dominant variable, which means that taking into account the more complex model with nonlinear form of the spline functions with interaction effects, all three dependent variables are significant in the modeled relationship. When CC_i is removed from model 2, the R^2 statistic is reduced from 0.7214 to 0.5946, while the *GCV* value increases from 0.0048 to 0.0067. It can therefore be concluded that the classification of the company into a specific group due to its central position in the VMN network is a significant indicator of the stock's return.

Finally, it should be noted that the obtained results of both models are the best for the penalty factor equal to 2 (most often in MARS approach the penalty factor is set to 2 or 3).

OLS regression analysis was performed for comparison. The results of the linear regression estimation are shown in Table 7.

Model 3 contains the same variables as Model 1, with no interactions between the variables. All variables are statistically significant at the level p = 0.001, and the regression coefficient for CC_i and $SVMB_i$ is positive, while for $MV_{(t-1)}$ it is negative. The linear model (Model 3) explains the variability of the dependent variable to a lesser extent compared to the MARS model (Model 1), adjusted R^2 are 0.5065 and 0.5856, respectively. However, the linear model confirms a statistically significant influence of the cluster of centrality on stock's return.

For a linear model, the importance of the predictors can be assessed by maximizing the absolute value of the standardized regression coefficient $|\beta|$. The order of importance of the variables is as follows (in parentheses $|\beta|$): 1) CC_i (0.624); 2) $SVMB_i$ (0.334); $MV_{(t-1)}$ (0.284). This confirms the conclusion of the significance of the cluster of centrality, drawn from the MARS model analysis.

Model 4 is a linear representation of the MARS model 2, where the individual independent variables and their interactions are included–(i) $MV_{(t-1)}*CC_i$; (ii) $SVMB_i^*CC_i$; (iii) $SVMB_i^*MV_{(t-1)}$. However, only 3 variables are statistically significant, including the CC_i variable (p<0.001), for which the regression coefficient is positive. The linear model 4 offers less ability to explain the variability than the corresponding MARS model 2, with the adjusted R^2 at 0.5566 and 0.7214, respectively. Hence, MARS regression models are more robust than linear models. Moreover, model 4 displays the collinearity problem of independent variables, which is indicated by a considerably exceeded acceptable level of the mean and maximum value of variance inflation factor (VIF>10).

5. Additional analysis

The limitation of the VMN built for 498 stocks is the imperfect projection of the full complexity for such a number of companies in the adopted one-year period. The value migration process should be analyzed over a properly defined period. It should not be too short in order to eliminate short-term fluctuations in stock prices. However, it should not be too long either to capture the scale of the process of changing the value of companies. The adopted period of one year results from the need to capture the current value migration process. In this section, the robustness of obtained results is carried out to narrow the research sample to the 250 largest companies from the S&P500 index. A similar approach was used in the study [110] to investigate the statistical properties of price returns for the 100 largest companies on the NYSE in one year, 2002. The aim of the analysis is to assess whether the above limitation biases the obtained results.

The largest companies are considered in terms of market capitalization in 2019. The analysis period remains the same; therefore, the dataset consists of 253 daily closing prices in every 250 time series. The value migration network for the 250 largest companies is presented in Fig A1, available in <u>S1 File</u>. The largest component of the VMN contains 219 firms.

The obtained topological layer of the SOM with the assumed dimensions of 3 x 4 again contains eight disjoint groups. The number of companies assigned to individual clusters is presented in Table A1 and Fig A2a in S1 File (see S1 File). The calculation of the cluster of centrality (CC_i) variable is provided in Table A2 in S1 File (see S1 File.).

The results of the MARS regression without the interaction of variables (model 5) are presented in <u>Table 8</u>. The final regression equation of model 5 is as follows:

 $\begin{aligned} ARRS_i &= 0.221 + 0.085 \cdot \max(0, \ CC_i - 0.714) - 0.264 \cdot \max(0, \ 0.714 - CC_i) + 1.747 \cdot \\ \max(0, \ SVMB_i - 0.033) - 3.504 \cdot \max(0, \ 0.033 - SVMB_i) - 0.227 \cdot \max(0, \ MV_{(t-1)} - 0.076) \ (43) \\ &+ 1.925 \cdot \max(0, \ 0.076 - MV_{(t-1)}) - 1.267 \cdot \max(0, \ SVMB_i - 0.064) \end{aligned}$

The graphical presentation of model 5 is demonstrated in Fig A3 in S1 File.

Model 5 is the counterpart of Model 1. By comparing both models, it can be seen that the structure of all the spline basis functions and the signs (positive/negative) of the regression coefficients are preserved. Furthermore, all regression coefficients of the contained spline basis functions are statistically significant (p<0.05). The value of the knot point of BF1 and BF2 is the same as in the original model: the value of 0.714 normalized knot; the value of 5 CC_i variable. This means that the critical threshold of the cluster for the non-linear effect remains unchanged. However, due to the change in the research sample, the knot point values have changed for the remaining functions related to the *SVMB_i* and $MV_{(t-1)}$ variables.

The results of Model 6 with second-order interactions between predictors are shown in Table 9.

Model 6 is the counterpart of Model 2. It consists of fewer base functions, six functions related to three variables, and one function with a second-order interaction. The following basic functions of model 6 (BF1 \div BF7) are equivalent to the basic functions of BF1, BF2, BF3, BF4, BF9, BF10, and BF8 of model 2, respectively. The construction of all basis functions and the signs of the regression coefficients are preserved and statistically significant at the significance level of 5%. In this case, the value of the knot point of BF1 and BF2 is the same as in the original model 2 and for models 1 and 5.

Table 8. Parameters and results for model 5.

MODEL 5								
	Parameters	Value	Value					
Dependent varial	ble	ARRS _i						
Independent vari	ables	CC_{i} ; $SVMB_{i}$; $MV_{(t-1)}$						
Maximum numb	er of basis functions	14						
Order of interact	ion	1						
Penalty		2						
Threshold		0.0005						
Removal of irrele	vant basis functions	Yes						
Number of observations		219	219					
Basis function		β	t statistics	p				
	Constant	0.221 (0.013)	16.854***	0.0000				
BF1	$\max(0, CC_i - 0.714)$	0.085 (0.063)	1.343*	0.0487				
BF2	$\max(0, 0.714 - CC_i)$	-0.264 (0.042)	-5.834***	0.0000				
BF3	$\max(0, SVMB_i - 0.033)$	1.747 (0.666)	2.622**	0.0094				
BF4	$\max(0, 0.033 - SVMB_i)$	-3.504 (0.408)	-8.592***	0.0000				
BF5	$\max(0, MV_{(t-1)} - 0.076)$	-0.227 (0.061)	-3.696***	0.0003				
BF6	$\max(0, 0.076 - MV_{(t-1)})$	1.925 (0.298)	6.466***	0.0000				
BF7	$max(0, SVMB_i - 0.064)$	-1.267 (0.675)	-1.877*	0.0491				
Results		Value	Value					
GCV		0.0061	0.0061					
R^2		0.6755	0.6755					
Adjusted R ²		0.6631	0.6631					

standard error in parentheses

 $^{\ast\ast\ast\ast},$ ** and * denote significant levels at 0.1%, 1%, and 5%, respectively

https://doi.org/10.1371/journal.pone.0276567.t008

The final regression equation of model 5 is as follows:

 $ARRS_{i} = 0.240 + 0.085 \cdot \max(0, CC_{i} - 0.714) - 0.169 \cdot \max(0, 0.714 - CC_{i}) + 0.710 \cdot \max(0, SVMB_{i} - 0.033) - 6.612 \cdot \max(0, 0.033 - SVMB_{i}) - 0.331 \cdot \max(0, MV_{(t-1)} - 0.076) + 2.776 \cdot \max(0, 0.076 - MV_{(t-1)}) + 19.963 \cdot \max(0, MV_{(t-1)} - 0.027) \cdot \max(0, 0.033 - SVMB_{i})$ (44)

Theraphic presentation of MARS model 6 is shown in Fig A4 in S1 File.

In summary, the robustness checks carried out demonstrated that the main findings are robust to the revised research sample in the form of a limitation to the 250 largest companies in the S&P500 index. Value migration can be analyzed in relation to a specified system of enterprises, e.g., the entire market, its selected part, or industry. The results obtained for the group of 250 largest companies are consistent with those for all companies from the S&P500 index. However, this does not mean that the value migration occurs separately within one of the two groups, the largest or the smallest companies. In other words, it is possible that the value flows between a company assigned to the TOP 250 group and smaller ones. However, the VMN built on the basis of all 498 stocks covers a wider range of value migration in the financial market.

6. Conclusions

In this work, a complex network has been formed using data for 498 companies from S&P500 index from 31 December 2018 to 31 December 2019. The value migration network

Table 9. Parameters and results for model 6.

MODEL 6							
	Parameters	Value	Value				
Dependent va	riable	ARRS _i					
Independent v	variables	CC _i ; SVMB _i ; MV _(t-1)					
Maximum nu	mber of basis functions	14					
Order of inter	action	2					
Penalty		2					
Threshold		0.0005					
Removal of irr	relevant basis functions	Yes					
Number of observations		219	219				
Basis function	1	β	t statistics	p			
	Constant	0.240 (0.015)	14.511***	0.0000			
BF1	$\max(0, CC_i - 0.714)$	0.085 (0.064)	1.474*	0.0492			
BF2	$\max(0, 0.714 - CC_i)$	-0.169 (0.042)	-6.105***	0.0000			
BF3	$\max(0, SVMB_i - 0.033)$	0.710 (0.152)	4.528***	0.0000			
BF4	$\max(0, 0.033 - SVMB_i)$	-6.612 (0.384)	-9.887***	0.0000			
BF5	$\max(0, MV_{(t-1)} - 0.076)$	-0.331 (0.083)	-3.732***	0.0002			
BF6	$\max(0, 0.076 - MV_{(t-1)})$	2.776 (0.363)	6.048***	0.0000			
BF7	$\max(0, 0.033 - SVMB_i) \cdot \max(0, MV_{(t-1)} - 0.027)$	19.963 (0.379)	1.858*	0.0376			
Results		Value					
GCV		0.0051					
R^2		0.7273	0.7273				
Adjusted R ²		0.7170					

standard error in parentheses

****, ** and * denote significant levels at 0.1%, 1%, and 5%, respectively

https://doi.org/10.1371/journal.pone.0276567.t009

constructed with the use of MST-Pathfinder has been employed as a tool to analyze the relationship between VM process and stock returns. Specifically, a complex network, artificial neutral network, and MARS regression have been applied to investigate the relationship between centrality measures of the VMN and individual stock returns. This paper is the first study that incorporates the above three approaches in the analysis of real complex financial systems.

This study explored the relationship between self-organizing feature map representing the centrality of a company's position in the VMN, and stock's return. Specifically, the investigation was focused on whether the central position of the company in the VMN affects the annualized rate of return on shares. The results obtained from the empirical analysis show that the topological position in the VMN has a statistically significant effect on the stock's returns. This study appears to be the first to demonstrate that value migration, measured by means of a self-organizing map based on the centrality measures of the VMN, plays an important role in determining stock's return levels.

Both MARS regression models show a non-linear effect of the company's centrality position in the VMN operationalized by the SOM on the annualized rate of return on shares. For firms classified into neurons (3,2); (3,1), and (2,1)–with CCi at 6, 7, or 8 –the company's place on the self-organizing feature map in terms of its central position in the VMN has a positive effect on stock's return and for companies from neurons (2,4); (1;4); (3,4); (1,3)–with CCi at 1, 2, 3, or 4 –a negative effect is demonstrated. However, the absolute values of the regression coefficients of the basis functions are different, which means the diverse strength of the influence of the variable centrality of cluster (CC_i) on the stock's return. In conclusion, the results of this study reveal that the level of centrality in the VMN is a non-linear indicator of the annualized rate of return on shares. This is confirmed by worse results in explaining the variability of the dependent variable by comparable linear models.

This study contributes to the area of (i) complex network; (ii) portfolio optimization, (iii) value-based management. Firstly, these results emphasize the importance of financial filtered networks and the position of companies in the value migration network. The place of the enterprise on the SOM is not only a statistically significant factor affecting the annualized rate of return on shares, but also shows high relative importance in the MARS regression models. For model 1, the classification of the firm into a cluster of network centrality is the most important determinant of the dependent variable. For model 2, the relative importance of clusters according to network centrality is over 53% of the most dominant variable.

Secondly, in the context of building an optimal stock portfolio strategy, it cannot be clearly indicated whether it is better to invest in central or peripheral stocks. The resulting classification using the Kohonen network and the four centrality measures does not determine whether the analyzed assets are central or peripheral. However, previous study [59] has revealed the diversifying impact of the centrality level in VMN on the rate of return on shares. This is due to the different nature of the individual centrality measures used, for which a positive effect was demonstrated for in-degree centrality, and a negative one for out-degree and entropy centrality. In other words, depending on the different type of centrality measures, better results in terms of the rate of return on shares are achieved by central or peripheral companies. It should be pointed out that the obtained results cannot be directly compared with the results in terms of the level of stock's return achieved by a set of stocks or a portfolio consisting of central or peripheral assets. Previous studies have been conducted with the use of different centrality measures dedicated to undirected networks (degree, closeness, betweenness, eigenvector centrality), which is the cross-correlation of log-return of stock price network. However, SOM based on VMN centrality can be useful in constructing the optimal portfolio.

Thirdly, one potential application of this work is to support value-based management evaluation tools. The topological position in the value migration network has an opportunity to become a new assessment measure in terms of the efficiency of implementing the concept of value-based management. Companies capable of intercepting the value in the financial market will be ranked higher on the SOM of network centrality. Consequently, enterprises in the inflow phase achieve higher rates of return on shares, providing a reliable view on the achievement of the company's goal of maximizing the value of the enterprise.

Future studies should focus on the impact of the type of sector on the VM process. Value may flow between separate companies, but the process of value fluctuation can be considered at a higher level of aggregation, i.e., economic sectors. Further research is needed to investigate value migration in a dynamic approach, where the subject of the study is a dynamic network. Such a network pattern will enable the analysis of the intensity of changes in the value migration process. However, parameters such as the window length and the rolling step must be properly set to use the rolling window approach for the value migration network.

Supporting information

S1 File. Supporting information for: The effect of self-organizing map architecture based on the value migration network centrality measures on stock return. Evidence from the US market. The Supporting Information, S1 File, contains additional tables and graphs for the robustness analysis.

(PDF)

S1 Dataset. (XLSX)

Author Contributions

Conceptualization: Dariusz Siudak.

Data curation: Dariusz Siudak.

Formal analysis: Dariusz Siudak.

Investigation: Dariusz Siudak.

Methodology: Dariusz Siudak.

Validation: Dariusz Siudak.

Visualization: Dariusz Siudak.

Writing - original draft: Dariusz Siudak.

Writing - review & editing: Dariusz Siudak.

References

- 1. Siudak D. The theoretical links between enterprise value migration, capital migration, and capital commitments' migration. Acta Scientiarum Polonorum Oeconomia. 2015; 14: 153–162.
- Strogatz SH. Exploring complex networks. Nature. 2001. pp. 268–276. <u>https://doi.org/10.1038/</u> 35065725 PMID: 11258382
- Albert R, Barabási AL. Statistical mechanics of complex networks. Rev Mod Phys. 2002; 74: 47–97. https://doi.org/10.1103/RevModPhys.74.47
- 4. Newman MEJ. Power laws, Pareto distributions and Zipf's law. Contemp Phys. 2005; 46: 323–351. https://doi.org/10.1080/00107510500052444
- Song C, Havlin S, Makse HA. Self-similarity of complex networks. Nature. 2005; 433: 392–395. <u>https://doi.org/10.1038/nature03248 PMID: 15674285</u>
- Borgatti SP. Centrality and network flow. Soc Networks. 2005; 27: 55–71. https://doi.org/10.1016/j. socnet.2004.11.008
- 7. Newman MEJ. Networks: An Introduction. Oxford: Oxford University Press; 2010.
- 8. Barabasi A-L. Network science. Cambridge University Press; 2016.
- Battiston S, Caldarelli G, May RM, Roukny T, Stiglitz JE. The price of complexity in financial networks. Proc Natl Acad Sci U S A. 2016; 113: 10031–10036. https://doi.org/10.1073/pnas.1521573113 PMID: 27555583
- Cimini G, Squartini T, Saracco F, Garlaschelli D, Gabrielli A, Caldarelli G. The statistical physics of real-world networks. Nature Reviews Physics. 2019. pp. 58–71. https://doi.org/10.1038/s42254-018-0002-6
- Mantegna RN, Stanley HE. An Introduction to Econophysics. Correlation and Complexity in Finance. Cambridge University Press; 1999. https://doi.org/10.1017/cbo9780511755767
- Mantegna RN. Hierarchical structure in financial markets. European Physical Journal B. 1999; 11: 193–197. https://doi.org/10.1007/s100510050929
- Onnela JP, Chakraborti A, Kaski K, Kertész J, Kanto A. Asset Trees and Asset Graphs in Financial Markets. Physica Scripta T. 2003. pp. 48–54. https://doi.org/10.1238/physica.topical.106a00048
- 14. Onnela JP, Kaski K, Kertész J. Clustering and information in correlation based financial networks. European Physical Journal B. 2004. pp. 353–362. https://doi.org/10.1140/epjb/e2004-00128-7
- 15. Bonanno G, Lillo F, Mantegna RN. High-frequency cross-correlation in a set of stocks. Quant Finance. 2001; 1: 96–104. https://doi.org/10.1080/713665554
- Kenett DY, Tumminello M, Madi A, Gur-Gershgoren G, Mantegna RN, Ben-Jacob E. Dominating clasp of the financial sector revealed by partial correlation analysis of the stock market. PLoS One. 2010; 5: 1–14. https://doi.org/10.1371/journal.pone.0015032 PMID: 21188140

- Chu J, Nadarajah S. A statistical analysis of UK financial networks. Physica A: Statistical Mechanics and its Applications. 2017; 471: 445–459. https://doi.org/10.1016/j.physa.2016.12.073
- Tang Y, Xiong JJ, Jia ZY, Zhang YC. Complexities in financial network topological dynamics: Modeling of emerging and developed stock markets. Complexity. 2018; 2018: 1–31. <u>https://doi.org/10.1155/ 2018/4680140</u>
- 19. Isogai T. Dynamic correlation network analysis of financial asset returns with network clustering. Appl Netw Sci. 2017; 2: 1–30. https://doi.org/10.1007/s41109-017-0031-6 PMID: 30443563
- Podobnik B, Horvatic D, Petersen AM, Stanley HE. Cross-correlations between volume change and price change. Proc Natl Acad Sci U S A. 2009; 106: 22079–22084. <u>https://doi.org/10.1073/pnas.0911983106</u> PMID: 20018772
- Schuenemann JH, Ribberink N, Katenka N. Japanese and Chinese Stock Market Behaviour in Comparison–an analysis of dynamic networks. Asia Pacific Management Review. 2020; 25: 99–110. https://doi.org/10.1016/j.apmrv.2019.10.002
- Pharasi HK, Sharma K, Chatterjee R, Chakraborti A, Leyvraz F, Seligman TH. Identifying long-term precursors of financial market crashes using correlation patterns. New J Phys. 2018; 20: 1–11. <u>https:// doi.org/10.1088/1367-2630/aae7e0</u>
- Sultornsanee S, Radhakrishnan S, Falco D, Zeid A, Kamarthi S. Phase synchronization approach to construction and analysis of stock correlation network. Procedia Computer Science. 2011. pp. 52–56. https://doi.org/10.1016/j.procs.2011.08.012
- 24. Brida JG, Matesanz D, Seijas MN. Network analysis of returns and volume trading in stock markets: The Euro Stoxx case. Physica A: Statistical Mechanics and its Applications. 2016; 444: 751–764. https://doi.org/10.1016/j.physa.2015.10.078
- Birch J, Pantelous AA, Soramäki K. Analysis of Correlation Based Networks Representing DAX 30 Stock Price Returns. Comput Econ. 2016; 47: 501–525. https://doi.org/10.1007/s10614-015-9481-z
- Zeng ZJ, Xie C, Yan XG, Hu J, Mao Z. Are stock market networks non-fractal? Evidence from New York Stock Exchange. Financ Res Lett. 2016; 17: 97–102. https://doi.org/10.1016/j.frl.2016.02.002
- Zhao L, Li W, Cai X. Structure and dynamics of stock market in times of crisis. Physics Letters, Section A: General, Atomic and Solid State Physics. 2016; 380: 654–666. <u>https://doi.org/10.1016/j.physleta.</u> 2015.11.015
- Tola V, Lillo F, Gallegati M, Mantegna RN. Cluster analysis for portfolio optimization. J Econ Dyn Control. 2008; 32: 235–258. https://doi.org/10.1016/j.jedc.2007.01.034
- Boginski V, Butenko S, Shirokikh O, Trukhanov S, Gil Lafuente J. A network-based data mining approach to portfolio selection via weighted clique relaxations. Ann Oper Res. 2014; 216: 23–34. https://doi.org/10.1007/s10479-013-1395-3
- 30. Tumminello M, Aste T, di Matteo T, Mantegna RN. A tool for filtering information in complex systems. Proc Natl Acad Sci U S A. 2005; 102: 10421–10426. <u>https://doi.org/10.1073/pnas.0500298102</u> PMID: 16027373
- Tumminello M, Lillo F, Mantegna RN. Correlation, hierarchies, and networks in financial markets. J Econ Behav Organ. 2010; 75: 40–58. https://doi.org/10.1016/j.jebo.2010.01.004
- Pozzi F, Di Matteo T, Aste T. Spread of risk across financial markets: Better to invest in the peripheries. Sci Rep. 2013; 3: 1–7. https://doi.org/10.1038/srep01665 PMID: 23588852
- Eom C, Park JW. Effects of common factors on stock correlation networks and portfolio diversification. International Review of Financial Analysis. 2017; 49: 1–11. https://doi.org/10.1016/j.irfa.2016.11.007
- Haluszczynski A, Laut I, Modest H, Räth C. Linear and nonlinear market correlations: Characterizing financial crises and portfolio optimization. Phys Rev E. 2017;96. <u>https://doi.org/10.1103/PhysRevE.</u> 96.062315 PMID: 29347332
- Millington T, Niranjan M. Construction of minimum spanning trees from financial returns using rank correlation. Physica A: Statistical Mechanics and its Applications. 2021;566. <u>https://doi.org/10.1016/j.physa.2020.125605</u>
- Todorova Z. Firm returns and network centrality. Risk Governance and Control: Financial Markets and Institutions. 2019; 9: 74–82. https://doi.org/10.22495/rgcv9i3p6
- Eng-Uthaiwat H. Stock market return predictability: Does network topology matter? Review of Quantitative Finance and Accounting. 2018; 51: 433–460. https://doi.org/10.1007/s11156-017-0676-3
- Hu S, Gu Z, Wang Y, Zhang X. An analysis of the clustering effect of a jump risk complex network in the Chinese stock market. Physica A: Statistical Mechanics and its Applications. 2019; 523: 622–630. https://doi.org/10.1016/j.physa.2019.01.114
- Wu F, Zhang D, Zhang Z. Connectedness and risk spillovers in China's stock market: A sectoral analysis. Economic Systems. 2019; 43: 1–14. https://doi.org/10.1016/j.ecosys.2019.100718

- Wang GJ, Xie C, Chen S. Multiscale correlation networks analysis of the US stock market: a wavelet analysis. J Econ Interact Coord. 2017; 12: 561–594. https://doi.org/10.1007/s11403-016-0176-x
- Zhao L, Wang GJ, Wang M, Bao W, Li W, Stanley HE. Stock market as temporal network. Physica A: Statistical Mechanics and its Applications. 2018; 506: 1104–1112. <u>https://doi.org/10.1016/j.physa.</u> 2018.05.039
- 42. Brida JG, Risso WA. Hierarchical structure of the German stock market. Expert Syst Appl. 2010; 37: 3846–3852. https://doi.org/10.1016/j.eswa.2009.11.034
- Vandewalle N, Brisbois F, Tordoir X. Non-random topology of stock markets. Quant Finance. 2001; 1: 372–374. https://doi.org/10.1088/1469-7688/1/3/308
- 44. Lee KE, Lee JW, Hong BH. Complex networks in a stock market. Comput Phys Commun. 2007; 177: 186. https://doi.org/10.1016/j.cpc.2007.02.047
- 45. Tse CK, Liu J, Lau FCM. A network perspective of the stock market. J Empir Finance. 2010; 17: 659–667. https://doi.org/10.1016/j.jempfin.2010.04.008
- Boginski V, Butenko S, Pardalos PM. Mining market data: A network approach. Comput Oper Res. 2006; 33: 3171–3184. https://doi.org/10.1016/j.cor.2005.01.027
- Kim K, Kim SY, Ha DH. Characteristics of networks in financial markets. Comput Phys Commun. 2007; 177: 184–185. https://doi.org/10.1016/j.cpc.2007.02.037
- Liu J, Tse CK, He K. Fierce stock market fluctuation disrupts scalefree distribution. Quant Finance. 2011; 11: 817–823. https://doi.org/10.1080/14697680902991627
- Eom C, Oh G, Kim S. Statistical investigation of connected structures of stock networks in a financial time series. Journal of the Korean Physical Society. 2008. pp. 3837–3841. <u>https://doi.org/10.3938/jkps.53.3837</u>
- Eom C, Oh G, Jung WS, Jeong H, Kim S. Topological properties of stock networks based on minimal spanning tree and random matrix theory in financial time series. Physica A: Statistical Mechanics and its Applications. 2009; 388: 900–906. https://doi.org/10.1016/j.physa.2008.12.006
- An F, Gao X, Guan J, Li H, Liu Q. An evolution analysis of executive-based listed company relationships using complex networks. Physica A: Statistical Mechanics and its Applications. 2016; 447: 276– 285. https://doi.org/10.1016/j.physa.2015.12.050
- Namaki A, Shirazi AH, Raei R, Jafari GR. Network analysis of a financial market based on genuine correlation and threshold method. Physica A: Statistical Mechanics and its Applications. 2011; 390: 3835–3841. https://doi.org/10.1016/j.physa.2011.06.033
- Mai Y, Chen H, Meng L. An analysis of the sectorial influence of CSI300 stocks within the directed network. Physica A: Statistical Mechanics and its Applications. 2014; 396: 235–241. <u>https://doi.org/10.1016/j.physa.2013.11.016</u>
- 54. Yan XG, Xie C, Wang GJ. Stock market network's topological stability: Evidence from planar maximally filtered graph and minimal spanning tree. Int J Mod Phys B. 2015;29. <u>https://doi.org/10.1142/S0217979215501611</u>
- Garas A, Argyrakis P. Correlation study of the Athens Stock Exchange. Physica A: Statistical Mechanics and its Applications. 2007; 380: 399–410. https://doi.org/10.1016/j.physa.2007.02.097
- Nguyen Q, Nguyen NKK, Nguyen LHN. Dynamic topology and allometric scaling behavior on the Vietnamese stock market. Physica A: Statistical Mechanics and its Applications. 2019; 514: 235–243. https://doi.org/10.1016/j.physa.2018.09.061
- 57. Barbi AQ, Prataviera GA. Nonlinear dependencies on Brazilian equity network from mutual information minimum spanning trees. Physica A: Statistical Mechanics and its Applications. 2019; 523: 876–885. https://doi.org/10.1016/j.physa.2019.04.147
- Tu C. Cointegration-based financial networks study in Chinese stock market. Physica A: Statistical Mechanics and its Applications. 2014; 402: 245–254. https://doi.org/10.1016/j.physa.2014.01.071
- **59.** Siudak D. A network analysis of the value migration process on the financial market. The effect of value migration network structure on stock returns. Expert Syst Appl. 2022; 191: 116129. <u>https://doi.org/10.1016/j.eswa.2021.116129</u>
- Chen K, Luo P, Sun B, Wang H. Which stocks are profitable? A network method to investigate the effects of network structure on stock returns. Physica A: Statistical Mechanics and its Applications. 2015; 436: 224–235. https://doi.org/10.1016/j.physa.2015.05.047
- Qiao H, Xia Y, Li Y. Can network linkage effects determine return? Evidence from Chinese stock market. PLoS One. 2016; 11: 1–25. https://doi.org/10.1371/journal.pone.0156784 PMID: 27257816
- Huang WQ, Yao S, Zhuang XT, Yuan Y. Dynamic asset trees in the US stock market: Structure variation and market phenomena. Chaos Solitons Fractals. 2017; 94: 44–53. https://doi.org/10.1016/j. chaos.2016.11.007

- Kaya H. Eccentricity in asset management. The Journal of Network Theory in Finance. 2015; 1. https://doi.org/10.21314/jntf.2015.003
- Ma J, Yang J, Zhang X, Huang Y. Analysis of Chinese stock market from a complex network perspective: Better to invest in the central. Chinese Control Conference, CCC. 2015. pp. 8606–8611. <u>https:// doi.org/10.1109/ChiCC.2015.7261003</u>
- Huang C, Zhao X, Su R, Yang X, Yang X. Dynamic network topology and market performance: A case of the Chinese stock market. International Journal of Finance and Economics. 2020; 1–17. <u>https://doi.org/10.1002/ijfe.2253</u>
- Onnela JP, Chakraborti A, Kaski K, Kertész J, Kanto A. Dynamics of market correlations: Taxonomy and portfolio analysis. Phys Rev E Stat Phys Plasmas Fluids Relat Interdiscip Topics. 2003;68. https://doi.org/10.1103/PhysRevE.68.056110 PMID: 14682849
- **67.** Zhao L, Li W, Fenu A, Podobnik B, Wang Y, Stanley HE. The q-dependent detrended cross-correlation analysis of stock market. Journal of Statistical Mechanics: Theory and Experiment. 2018; 2018: 1–28. https://doi.org/10.1088/1742-5468/aa9db0
- Li Y, Jiang XF, Tian Y, Li SP, Zheng B. Portfolio optimization based on network topology. Physica A: Statistical Mechanics and its Applications. 2019; 515: 671–681. https://doi.org/10.1016/j.physa.2018. 10.014
- Onnela JP, Chakraborti A, Kaski K, Kertész J. Dynamic asset trees and portfolio analysis. European Physical Journal B. 2002; 30: 285–288. https://doi.org/10.1140/epjb/e2002-00380-9
- Peralta G, Zareei A. A network approach to portfolio selection. J Empir Finance. 2016; 38: 157–180. https://doi.org/10.1016/j.jempfin.2016.06.003
- 71. Ren F, Lu YN, Li SP, Jiang XF, Zhong LX, Qiu T. Dynamic portfolio strategy using clustering approach. PLoS One. 2017; 12: 1–23. https://doi.org/10.1371/journal.pone.0169299 PMID: 28129333
- Millington T, Niranjan M. Stability and similarity in financial networks—How do they change in times of turbulence? Physica A: Statistical Mechanics and its Applications. 2021; 574. https://doi.org/10.1016/j. physa.2021.126016
- 73. Miccichè S, Bonanno G, Lillo F, Mantegna RN. Degree stability of a minimum spanning tree of price return and volatility. Physica A: Statistical Mechanics and its Applications. 2003. pp. 66–73. <u>https://doi.org/10.1016/S0378-4371(03)00002-5</u>
- 74. Bonanno G, Caldarelli G, Lillo F, Miccichè S, Vandewalle N, Mantegna RN. Networks of equities in financial markets. European Physical Journal B. 2004. pp. 363–371. <u>https://doi.org/10.1140/epjb/e2004-00129-6</u>
- 75. Bonanno G, Caldarelli G, Lillo F, Mantegna RN. Topology of correlation-based minimal spanning trees in real and model markets. Phys Rev E Stat Phys Plasmas Fluids Relat Interdiscip Topics. 2003; 68. https://doi.org/10.1103/PhysRevE.68.046130 PMID: 14683025
- 76. Coelho R, Hutzler S, Repetowicz P, Richmond P. Sector analysis for a FTSE portfolio of stocks. Physica A: Statistical Mechanics and its Applications. 2007; 373: 615–626. <u>https://doi.org/10.1016/j.physa.2006.02.050</u>
- Borghesi C, Marsili M, Miccichè S. Emergence of time-horizon invariant correlation structure in financial returns by subtraction of the market mode. Phys Rev E Stat Nonlin Soft Matter Phys. 2007;76. https://doi.org/10.1103/PhysRevE.76.026104 PMID: 17930101
- Brida JG, Risso WA. Dynamics and structure of the 30 largest North American companies. Comput Econ. 2010; 35: 85–99. https://doi.org/10.1007/s10614-009-9187-1
- 79. Kazemilari M, Djauhari MA. Correlation network analysis for multi-dimensional data in stocks market. Physica A: Statistical Mechanics and its Applications. 2015; 429: 62–75. https://doi.org/10.1016/j. physa.2015.02.052
- Coletti P. Comparing minimum spanning trees of the Italian stock market using returns and volumes. Physica A: Statistical Mechanics and its Applications. 2016; 463: 246–261. <u>https://doi.org/10.1016/j.physa.2016.07.029</u>
- Lee JW, Nobi A. State and Network Structures of Stock Markets Around the Global Financial Crisis. Comput Econ. 2018; 51: 195–210. https://doi.org/10.1007/s10614-017-9672-x
- Long W, Guan L, Shen J, Song L, Cui L. A complex network for studying the transmission mechanisms in stock market. Physica A: Statistical Mechanics and its Applications. 2017; 484: 345–357. https://doi. org/10.1016/j.physa.2017.04.043
- Yao H, Memon BA. Network topology of FTSE 100 Index companies: From the perspective of Brexit. Physica A: Statistical Mechanics and its Applications. 2019; 523: 1248–1262. https://doi.org/10.1016/ j.physa.2019.04.106
- Djauhari MA. A robust filter in stock networks analysis. Physica A: Statistical Mechanics and its Applications. 2012; 391: 5049–5057. https://doi.org/10.1016/j.physa.2012.05.060

- 85. Kravitz D. Two Comments on Minimum Spanning Trees. The Bulletin of the ICA. 2007; 49: 7–10.
- Djauhari MA, Gan SL. Optimality problem of network topology in stocks market analysis. Physica A: Statistical Mechanics and its Applications. 2015; 419: 108–114. <u>https://doi.org/10.1016/j.physa.2014</u>. 09.060
- 87. Quirin A, Cordón O, Guerrero-Bote VP, Vargas-Quesada B, Moya-Anegón F. A quick MST-based algorithm to obtain pathfinder networks (∞, n—1). Journal of the American Society for Information Science and Technology. 2008; 59: 1912–1924. https://doi.org/10.1002/asi.20904
- Kruskal JB. On the Shortest Spanning Subtree of a Graph and the Traveling Salesman Problem. Proceedings of the American Mathematical Society. 1956; 7: 48. https://doi.org/10.2307/2033241
- Kohonen T. Self-organized formation of topologically correct feature maps. Biol Cybern. 1982; 43: 59– 69. https://doi.org/10.1007/BF00337288
- 90. Kohonen T. Self-Organizing Maps, Springer series in Information Sciences. Springer-Verlag. 2001.
- Kohonen T. Essentials of the self-organizing map. Neural Networks. 2013; 37: 52–65. <u>https://doi.org/10.1016/j.neunet.2012.09.018 PMID: 23067803</u>
- 92. Haykin S. Neural Networks: A Comprehensive Foundation. Second edi. Prentice Hall International, Inc.; 1999.
- Zhang H, Xu X, Yan Y, Xu P, Lu Y, Hou Z. Recognition of Partial Discharge in Switchgear Based on Kohonen Network. 2020 IEEE Electrical Insulation Conference, EIC 2020. 2020. pp. 542–545. https:// doi.org/10.1109/EIC47619.2020.9158688
- 94. Friedman JH. Multivariate Adaptive Regression Splines. The Annals of Statistics. 1991; 19: 1–67. https://doi.org/10.1214/aos/1176347963
- **95.** Hastie Trevor, Tibshirani Robert, Friedman J. The Elements of Statistical Learning The Elements of Statistical LearningData Mining, Inference, and Prediction. Second Edi. Springer. 2009.
- 96. Craven P, Wahba G. Smoothing noisy data with spline functions—Estimating the correct degree of smoothing by the method of generalized cross-validation. Numer Math (Heidelb). 1979; 31: 377–403. https://doi.org/10.1007/BF01404567
- Friedman JH, Roosen CB. An introduction to multivariate adaptive regression splines. Stat Methods Med Res. 1995; 4: 197–217. https://doi.org/10.1177/096228029500400303 PMID: 8548103
- 98. Siudak D. Inter-Industrial Value Migration. Quantitative methods in economics. 2014; 15: 232-241.
- 99. Esmaeilpour Moghadam H, Mohammadi T, Feghhi Kashani M, Shakeri A. Complex networks analysis in Iran stock market: The application of centrality. Physica A: Statistical Mechanics and its Applications. 2019; 531: 1–13. https://doi.org/10.1016/j.physa.2019.121800
- Tutzauer F. Entropy as a measure of centrality in networks characterized by path-transfer flow. Soc Networks. 2007; 29: 249–265. https://doi.org/10.1016/j.socnet.2006.10.001
- Kleinberg JM. Authoritative sources in a hyperlinked environment. Journal of the ACM. 2011. pp. 604– 632. https://doi.org/10.1145/324133.324140
- 102. http://finance.yahoo.com. In: accessed on 10.03.2020.
- 103. https://www.refinitiv.com. In: accessed on 25.03.2020.
- 104. Fruchterman TMJ, Reingold EM. Graph drawing by force-directed placement. Softw Pract Exp. 1991; 21: 1129–1164. https://doi.org/10.1002/spe.4380211102
- 105. Cyram. NetMinerVersion 4.5.0. Seoul: Cyram Inc.; 2022. Available: http://www.netminer.com/
- 106. TIBCO Software Inc. Statistica (data analysis software system). 2017. Available: http://statistica.com
- 107. Tumminello M, di Matteo T, Aste T, Mantegna RN. Correlation based networks of equity returns sampled at different time horizons. European Physical Journal B. 2007; 55: 209–217. https://doi.org/10. 1140/epjb/e2006-00414-4
- Brida JG, Risso WA. Multidimensional minimal spanning tree: The Dow Jones case. Physica A: Statistical Mechanics and its Applications. 2008; 387: 5205–5210. https://doi.org/10.1016/j.physa.2008.05.009
- 109. Kwapień J, Drożdż S. Physical approach to complex systems. Physics Reports. 2012. pp. 115–226. https://doi.org/10.1016/j.physrep.2012.01.007
- 110. Coronnello C, Tumminello M, Lillo F, Miccichè S, Mantegna RN. Economic sector identification in a set of stocks traded at the New York Stock Exchange: a comparative analysis. Noise and Stochastics in Complex Systems and Finance. 2007. p. 66010T. https://doi.org/10.1117/12.729619